. . Cal Poly CPE 101: Fundamentals of Computer Science I Alexander Dekhtyar . . Programming Languages Alex’s Scorecard

Programming Languages

Note: Below is a list of programming languages in which I have written programs.

No. Language Origin of name Comments 1. Pascal Blaise Pascal first program: game of Life, punchtape 2. Basic self-explanatory high-school CS, Apple II clones 3. Prolog Programming in Logic Used for AI applications 4. C A, B, C Learned while at University of Alabama 5. Clarion ? Database development language 6. LISP List Processing Programming languages class, U of A 7. Awk Alfred Aho, T.A. in grad school for Peter Weinberger, Programming Languages class Brian Kernighan 8. ml metalanguage same as above 9. C++ ”++” is ”increment” in C learned (finally) in grad school 10. SQL Structured Query Language database query language 11. Java Coffee bean? Taught in a web programming course 12. Perl originally ”Pearl” (Parable of the Pearl) powerful script language for Unix 13. JavaScript Java+Script taught in web programming course 14. PHP Personal Home Page taught in web programming course 15. XQuery XML Query language taught in grad. databases class 16. Smodels Stable Models extension of Prolog for AI applications